Author’s response to reviews

Title: Child marriage among boys in high-prevalence countries: an analysis of sexual and reproductive health outcomes

Authors:

Christina Misunas (cmisunas@gmail.com)
Colleen Gastón (cmurray@unicef.org)
Claudia Cappa (ccappa@unicef.org)

Version: 4 Date: 22 Jul 2019

Author’s response to reviews:

Dear Editorial team,

To summarize the changes made:

We reformatted the bibliography so it was consistent with the citation style used by the production team.

We revised the title of the paper so it is grammatically correct and also more closely aligned with the content of the paper. The new title reads: Child marriage among boys in high-prevalence countries: an analysis of sexual and reproductive health outcomes

We made changes to the sentence structure in only two places for the sake of clarification.

We reviewed the tables to make sure the number of decimal points were consistent throughout.

We ask that the production team please capture the tables as fixed images as discussed in prior correspondence.

Many thanks,

Christina Misunas